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 Abstract 

 

Poly-3-hydroxybutyrate (PHB) is a biodegradable polymer which has physical properties 

similar to polypropylene & is having wide range of societal applications and more 

importantly medicinal applications. Numerous gram-positive and gram-negative 

microorganisms have the capability to accumulate Polyhydroxyalkanoates (PHAs) under 

nutrient limiting conditions. However, the PHB produced by gram-negative bacteria 

contains lipopolysaccharides (LPS) endotoxins in their outer membrane, which makes it 

unfit for medical applications whereas PHB produced by gram-positive bacterial lacks 

LPS hence making it more suitable for medical applications. Therefore, in the present 

investigations research effortswere directed towards production of medical grade 

biopolymers using gram positive bacteria (Bacillus thuringiensis IAM12077). 

The main objective of the present study was to establish PHB production process using 

Bacillus thuringiensis IAM12077 as a model organism. It was observed that B. 

thuringiensis was able to utilize glycerol as a major carbon source for PHB production, 

thereby indicating the possibility of utilization of inexpensive crude glycerol (as well) as 

key substrate, which is a by-product of the biofuel industry. To obtain maximum biomass 

PHB accumulation by B. thuringiensis, concentrations of different nutrients were 

statistically optimized. Batch cultivation in a 7L bioreactor using optimized medium 

recipe resulted in biomass concentration of 6.68 g/L with a PHB concentration of 2.71 

g/L in 33 hours. Possibility of substrate inhibition by selected major limiting nutrient 

(Carbon source; glycerol) was also explored which demonstrated a decrease in specific 

growth rate of B. thuringiensis beyond 15g/L with respect to glycerol. To further enhance 

the PHB yield and productivity, cultivation of B. thuringiensis was attempted in 7 L 

bioreactor using optimized glycerol and glucose concentration in the ratio of 1:1, (at a 

total concentration of 15g/L) along with statistically optimized medium components 

which exhibited a biomass and PHB concentration of 8.45 g/L and 3.36 g/L, respectively 
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in 33 hours. This was significantly higher as compared to bioreactor batch cultivation 

conducted with glycerol as sole carbon source. 

Successful attempts were then made to identify two different batch mathematical models, 

e.g., for glycerol as substrate and mixed substrate i.e. glycerol with glucose for production 

by B. thuringiensis IAM12077 using substrate inhibition studies & 7 L batch kinetics 

data. Both the models were then extrapolated to simulate different nutrient feeding 

strategies in fed-batch cultivation conditions which included constant feed rate of glycerol 

and maintenance of pseudo steady state of glycerol. It was observed that fed-batch 

cultivation using constant feed rate strategy of glycerol yielded PHB concentration of 

10.4 g/L and 4.15 g/L, respectively in 33 hours the fed-batch cultivation conducted with 

mixed substrate & under pseudo steady state with respect to glycerol exhibited maximum 

biomass and PHB concentration of 8.42 g/L and 3.93 g/L, respectively in 36 hours. 

However, fed-batch cultivation conducted with mixed substrates & at constant feed 

strategy of mixed substrates resulted in biomass and PHB accumulation of 10.20 g/L and 

4.57 g/L, in 27 hours respectively. On the other hand, the fed-batch cultivation strategy 

with feeding of mixed substrate (glucose and glycerol) to maintain pseudo steady state 

with respect to glycerol resulted in biomass and a reasonably high PHB accumulation of 

12.74 g/L and 5.48 g/L, respectively in 33 hours. 

The batch mathematical model for mixed substrate was used to develop an (integrated 

batch-fed-batch-continuous with cell retention) cultivation strategy to further enhance the 

biomass and PHB accumulation in the bioreactor. The experimental implementation of 

this model based unique cultivation strategy resulted in biomass and PHB accumulation 

of 24.1 g/L and 9.84 g/L, respectively, reported so far for B. thuringiensis IAM12077. 

The present studies also demonstrated the successful scale-up of batch PHB fermentation 

from 7L to 70L bioreactor using Pg/V criterion where in   biomass and PHB concentration 

of 6.68 g/L and 2.71 g/L, respectively was obtained in both bioreactor cultivations. 
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सामा᭠यसाराशं 

 

 

Poly-3-hydroxybutyrate (PHB) एक जैविनम्  ननी पॉलीमर ह ैिजसमᱶ पॉलीᮧोपाइलीन के समान 

भौितक गुण होते ह ᱹऔर इसकᳱ ᳞ापक ᮰ेणी के सामािजक अनुᮧयोगᲂ और अिधक मह᭜वपणूᭅ ᱨप से 

औषधीय अनुᮧयोग होत े ह।ᱹ कई ᮕाम-पॉिजᳯटव और ᮕाम-नेगेᳯटव सू᭯मजीवᲂ मᱶ पोषक त᭜वᲂ कᳱ 

सीिमत ि᭭थित के तहत पॉलीहाइᮟॉ᭍सीक᭨कने᭗स (PHAs) को संिचत करन े कᳱ ᭃमता होती ह।ै 

हालाᳰंक, ᮕाम-नेगेᳯटव ब᭍ैटीᳯरया ᳇ारा िनᳶमत PHB मᱶ उनके बाहरी िझ᭨ली मᱶ िलपोपॉलीसेकेराइड 

(LPS) एडंोटॉि᭍सन होते ह,ᱹ जो इसे िचᳰक᭜सा अनुᮧयोगᲂ के िलए अयो᭏य बनाता ह,ै जबᳰक ᮕाम-

पॉिजᳯटव ब᭍ैटीᳯरया ᳇ारा उ᭜पाᳰदत PHB मᱶ LPS (एल-िप-एस) उपि᭭थत नहᱭ होता ह ै िजसके 

कारण PHB मेिडकल अनᮧुयोगᲂ के िलए अिधक उपयुᲦ होता ह।ैइसिलए, वतᭅमान अ᭟ययन मᱶ 

ᮕामपॉिजᳯटव बै᭍टीᳯरया (बैिसलस थुᳳरिगनेिसस IAM12077) का उपयोग करके मेिडकल ᮕेड 

बायोपॉिलमर उ᭜पादन पर ᭟यान कᱶ ᳰᮤत ᳰकया गया था। 

वतᭅमान अ᭟ययन का म᭎ुय उ᳎े᭫ य बेिसलस थुᳳरिगनेिसस IAM12077 का एक मॉडल जीव के 

ᱨप मᱶ उपयोग करके PHB उ᭜पादन ᮧᳰᮓया ᭭थािपत करना था। यह दखेा गया ᳰक बी. ᭝यᳳुरगᱶिसस 

ि᭏लसरॉल को PHB उ᭜पादन के िलए एक ᮧमुख काबᭅन ᮲ोत के ᱨप मᱶ उपयोग करने मᱶ सᭃम था, 

िजसस ेमहगें कᲬे ि᭏लसरॉल (साथ ही) ᮧमुख स᭣सᮝेट के उपयोग कᳱ संभावना का सकेंत िमलता ह,ै जो 

जैव ᲊधन उ᳒ोग का एक उप-उ᭜पाद ह।ै । बी. ᭝यᳳूरगᱶिसस ᳇ारा अिधकतम बायोमास (PHB) 

पीएचबी (PHB) संचय ᮧा᳙ करन ेके िलए, िविभ᳖ पोषक त᭜वᲂ कᳱ सांᮤता को सांि᭎यकᳱय ᱨप स े

अनुकूिलत ᳰकया गया था।अनुकूिलत मा᭟यम नु᭭खा का उपयोग कर 7 लीटर बैच बायोᳯरए᭍टर मᱶ 

उपज (cultivation) से बायोमास सांᮤण 6.68 ᮕाम/लीटर ᱟआ एव ं 2.71 ᮕाम/लीटरकᳱ PHB 

एकाᮕता का 33 घंटो मᱶ उतपादन ᱟआ। स᭣सᮝेट इनिहिबटबी कᳱ ᮧमुख सीिमत पोषक त᭜व (काबᭅन 

᮲ोत; ि᭏लसरॉल) कᳱ संभावना भी पता लगाया गया था, जो ᳰक 15ᮕाम/लीटर ि᭏लसरॉल से परे बी. 
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᭝यᳳूरगᱶिसस कᳱ िविश᳥ वृि᳍ दर मᱶ कमी को ᮧदᳶशत करता ह।ै PHB कᳱ उपज और उ᭜पादकता को 

और बढ़ाने के िलए, सांि᭎यकᳱय ᱨप स ेअनुकूिलत म᭟यम घटकᲂ के साथ, 1: 1 के अनपुात मᱶ ि᭏लसरॉल 

और ᭏लूकोज सांᮤता का उपयोग करत ेᱟए बी. थुᳳरिगएि᭠सस कᳱ खेती का ᮧयास 7 लीटरबायोᳯरए᭍टर 

मᱶ ᳰकया गया। िजसमᱶ 33 घंटे मᱶ ᮓमशः 8.45 ᮕाम/लीटर और 3.36 ᮕाम/लीटर के बायोमास और 

पीएचबी एकाᮕता का ᮧदशᭅन ᳰकया गया। यह केवल काबᭅन ᮲ोत के ᱨप मᱶ ि᭏लसरॉल के साथ 

आयोिजत बायोᳯरए᭍टर बचै कᳱ खेती कᳱ तुलना मᱶ काफᳱ अिधक था। 

तब स᭣सᮝेट (ि᭏लसरॉल) और िमि᮰त काबᭅन स᭣सᮝेट (ि᭏लसरॉल और ᭏लूकोज) के ᱨप मᱶ 

ि᭏लसरॉल के िलए दो अलग-अलग बैच गिणतीय मॉडल, जैस,े स᭣सᮝेट और िमि᮰त स᭣सᮝेट के ᱨप मᱶ 

ि᭏लसरॉल के िलए यानी ि᭏लसरॉल के साथ ᭏लूकोज के साथ उ᭜पादन के िलए बी. थुᳳरिगएि᭠सस 

IAM12077 स᭣सᮝेट िनषेध अ᭟ययन और 7 लीटर बैच कैनेटी᭍स डटेा का उपयोग करत ेᱟएउ᭜पादन 

के िलए सफल ᮧयासᲂ कᳱ पहचान कᳱ गई। दोनᲂ मॉडलᲂ को तब अलग-अलग पोषक त᭜वᲂ कᳱ िखला 

रणनीितयᲂ का अनुकरण करने के िलए अितᳯरᲦ ᱨप से तैयार ᳰकया गया था, फᳱड-बैच कᳱ खेती कᳱ 

ि᭭थितयᲂ मᱶ िजसमᱶ ि᭏लसरॉल कᳱ िनरंतर फ़ᳱड दर और ि᭏लसरॉल के छ᳑ ि᭭थर अव᭭था का रखरखाव 

शािमल था। यह दखेा गया ᳰक ि᭏लसरॉल कᳱ िनरंतर फ़ᳱड दर रणनीित का उपयोग करके फᳱड-बैच 

(fed-batch)कᳱ खेती न े10.4 ᮕाम/लीटर और 4.15 ᮕाम/लीटर कᳱ PHB सांᮤता ᮧा᳙ कᳱ, 33 घंटे 

मᱶ िमि᮰त स᭣सᮝेट और छ᳑ रा᭔य के संबंध मᱶ आयोिजत फᳱड-बैच कᳱ खेती के संबंध मᱶ ि᭏लसरॉल न े

अिधकतम बायोमास और PHBसांᮤता ᮓमश: 8.42 ᮕाम/लीटर और 3.93 ᮕाम/लीटर, को 36 घंटे मᱶ 

ᮧदᳶशत ᳰकया। हालाᳰंक, िमि᮰त स᭣सᮝे᭗स और िमि᮰त स᭣सᮝे᭗स कᳱ िनरंतर फ़ᳱड रणनीित के साथ 

आयोिजत बैच बैच कᳱ खेती के पᳯरणाम᭭वᱨप ᮓमशः 27 घंटे मᱶ 10.20ᮕाम/लीटर और 

4.57ᮕाम/लीटर का बायोमास संचय होता ह।ै दसूरी ओर, िमि᮰त-बैच स᭣सᮝेट (᭏लूकोज और 

ि᭏लसरॉल) कᳱ फᳱड-बैच कᳱ खेती कᳱ रणनीित के साथ-साथ ि᭏लसरॉल के संबंध मᱶ छ᳑ ि᭭थर ि᭭थित 

बनाए रखने के िलए बायोमास और पᳯरणाम᭭वᱨपउᲬ PHB, ᮓमशः 12.74 ᮕाम/लीटर और 5.48 

ᮕाम/लीटर 33 घंटे मᱶ ᱟआ। 

ᳰक᭛वन के शोरब े (fermentation broth) मᱶ बायोमास और PHB संचय को और बढ़ाने के 

िलए, एक एकᳱकृत बैच-फेड-बैच-िनरंतर उतपादन कᳱ रणनीित िवकिसत मॉडल कᳱ मदद से िवकिसत 

कᳱ गई। ᭏लकूोज और अ᭠य सांि᭎यकᳱय ᱨप स ेअनुकूिलत मीिडया घटकᲂ के साथ ि᭏लसरॉल के संबंध मᱶ 

छ᳑ ि᭭थर अव᭭था मᱶ फेड-बैच कᳱ खेती के बाद बायोᳯरए᭍टर मᱶ िनरंतर मोड मᱶ ताजा पोषक त᭜वᲂ को 

िखलाने से जुड़ी इस अिभनव रणनीित को लाग ु ᳰकया गया। इस उतपादन कᳱ रणनीित से 24.1 
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ᮕाम/लीटर बायोमास एव ं 9.84 ᮕाम/लीटर पीएचबी (PHB) सांᳰᮤत ᮧा᳙ ᱟआ, िजसमᱶ PHB 

उ᭜पादकता ि᭭थर अव᭭था मᱶ 0.298 ᮕाम/लीटर.घंटे थी, जो ᳰक बी. थुᳳरिगएि᭠सस उतपादन कᳱ सबस े

उᱫम रणनीित का ᮧमाण ह।ै 

वतᭅमान अ᭟ययनᲂ ने Pg/V मानदडं का उपयोग करत ेᱟए 7लीटरसे 70लीटरबायोᳯरए᭍टर 

तक बैच PHB ᳰक᭛वन के सफल पैमान-ेअप का ᮧदशᭅन ᳰकया, जहा ंबायोपास और PHB एकाᮕता 

ᮓमशः6.68 ᮕाम/लीटर और 2.71 ᮕाम/लीटर,दोनᲂ बायोᳯरए᭍टर कᳱ खेती मᱶ ᮧा᳙ ᳰकया गया था। 
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